Author's response to reviews

Title: Therapeutic drug monitoring of nevirapine in saliva in Uganda using high performance liquid chromatography and a low cost thin-layer chromatography technique

Authors:

Mohammed Lamorde (mlamorde@idi.co.ug)
Quirine Fillekes (q.fillekes@cbg-meb.nl)
Kim Sigaloff (k.sigaloff@pharmaccess.org)
Cissy Kityo (ckityo@yahoo.com)
Allan Buzibye (abuzibye@idi.co.ug)
Joshua Kayiwa (joshua.kayiwa@gmail.com)
Concepta Merry (cmerry@tcd.ie)
Lillian Nakatudde-Katumba (lnakatudde@yahoo.com)
David Burger (D.Burger@akf.umcn.nl)
Tobias Rinke de Wit (tinkedewit@pharmaccess.org)

Version: 2 Date: 4 April 2014

Author's response to reviews:

Authors' contribution and acknowledgment sections are included in the revised manuscript